Author's response to reviews

Title: Time trends in leisure time physical activity and physical fitness in elderly people: 20 year follow-up of the spanish population national health survey (1987-2006)

Authors:

Domingo Palacios-Ceña (domingo.palacios@urjc.es)
Cristina Alonso-Blanco (cristina.alonso@urjc.es)
Rodrigo Jiménez-García (rjimenezgarcia@yahoo.es)
Pilar Carrasco-Garrido (pilar.carrasco@urjc.es)
Valentín Hernández-Barrera (valentin.hernandez@urjc.es)
Elena Pileño-Martínez (elena.pileno@urjc.es)
Cesar Fernández-de-las-Peñas (cesar.fernandez@urjc.es)

Version: 6 Date: 12 October 2011

Author's response to reviews: see over
Response letter Manuscript 1154589831562773.R4

Time trends in leisure time physical activity and physical fitness in elderly people: 20 year follow-up of the spanish population national health survey (1987-2006)

Domingo Palacios-Ceña; Cristina Alonso-Blanco; Rodrigo Jiménez; Valentín Hernández-Barrera; Pilar Carrasco-Garrido; Elena Pileño-Martinez; César Fernández-de-las-Peñas

We follow all Formatting changes.

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is "First name Initial Last name" (e.g. Joe F Bloggs)

Response:

- The authors’ name should be in that order.
- The order of the names must follow that order by indexing problems, All articles we have published, our names are written in this order.
- Also, in Spain, there are names which have one name and a complex (compound) surname formed by two parts.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above). Please remove 'Physical Activity And Fitness In Older People' from each page.

Response: Removed.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: Done.

Thank you.
Best regards
The authors.